Efficient Protocols for Multimedia Streams on WDMA Networks

Lixin Wang, Maode Ma, and Mounir Hamdi, Member, IEEE

Abstract—This paper introduces a new approach to integrate different types of medium access control (MAC) protocols into a single wavelength-division-multiplexing (WDM) network system. The WDM network is based on a passive star coupler, and the purpose of integrating different MAC protocols is to efficiently accommodate various types of multimedia traffic streams with different characteristics and quality of service demands. Our integrated MAC protocol is termed multimedia wavelength-division multiple-access (M-WDMA). Three types of multimedia traffic streams are considered in this paper: constant-bit-rate traffic and two classes of variable-bit-rate traffic. Accordingly, three tunable transmitters and one fixed home channel receiver are used in the design of each WDM node. The transmitters transmit the three types of multimedia traffic streams in a *pipeline* fashion so as to overcome the tuning time overhead and to support parallel transmissions of traffic streams that emerge simultaneously. We further incorporate a dynamic bandwidth allocation scheme that dynamically adjusts the portions of bandwidth occupied by the three types of traffic streams according to their demands. Consequently the M-WDMA protocol achieves high utilization and efficiently adapts to the demands of the multimedia streams so as to guarantee their QoS. The performance of the M-WDMA is evaluated through a simple analytical model and extensive discrete-event simulations. It is shown that the M-WDMA can satisfy the QoS requirements of various mixes of multimedia traffic streams even under very stringent requirements. Moreover, we show that the M-WDMA outperforms conventional MAC protocols for WDM networks. As a result, we expect M-WDMA to be a good multimedia MAC candidate protocol for future-generation WDM networks.

Index Terms—Analytical modeling, MAC protocols, multimedia streams, performance evaluation, WDM networks.

I. INTRODUCTION

T is estimated that a single optical fiber has a capacity of at least 30 THz in its low-loss region (1.2–1.6 μ m). This huge bandwidth can be used for data transmission well beyond the terabits per second (Tb/s). At the present time, however, only a fraction of this huge capacity can be used due to well known *optoelectronic* bottleneck problems. These problems can

Manuscript received May 7, 2003. A preliminary version of this paper has won the Best Paper Award among 150 accepted papers at the 12th International Conference on Information Networking, Tokyo, Japan, 1998.

L. Wang was with the Department of Computer Science, Hong Kong University of Science and Technology, Kowloon, Hong Kong. He is now with Trumptech Ltd., Shenzen, China.

M. Ma was with the Department of Computer Science, Hong Kong University of Science and Technology, Kowloon, Hong Kong. He is now with Nanyang Technological University, Singapore 639798.

M. Hamdi is with the Department of Computer Science, Hong Kong University of Science and Technology, Kowloon, Hong Kong (e-mail: hamdi@ cs.ust.hk).

Digital Object Identifier 10.1109/JLT.2003.816815

be reduced by dividing the total bandwidth of the fiber into a number of noninterfering channels. These channels are of relatively lower bandwidths but with the advantage that they operate at the peak speed of the electronic processing devices. This technique is known as *wavelength-division multiplexing* (WDM). In a WDM architecture, a transmitter and a receiver have to tie onto the same wavelength (i.e., channel) so that transmission can be carried out successfully. Due to the limitation of the number of transceivers at each node, the complete connectivity of a WDM network requires the use of tunable transceivers, which can tune their working wavelength across all channels in the network. Unfortunately, one has to make a compromise between the tuning time, which can range from tens of nanoseconds to hundreds of milliseconds, and the tuning range, which can be from a few channels to hundreds of channels. Using the state-of-the-art technology of optical fibers, fast tuning time implies small tuning range, and vice versa. From a networking perspective, it was shown that increasing the number of channels can be quite beneficial. As a result, dealing with the overhead that can be associated with the tuning time becomes a crucial design problem in the WDM networks [5].

Several topologies have been proposed for the WDM networks [17], [18], a popular one being the single-hop passive star-coupled topology. Each transmitter in this single-hop network broadcasts its data to all other nodes through a passive star *coupler*. The receivers may then filter out a coming packet at a certain wavelength for their proper reception. This single-hop network is also referred to as a broadcast-and-select network. However, this WDM network comes with various challenges, the most important challenge being the design of the medium access control (MAC) protocols for these networks. One reason for this challenge is that all higher layer services (e.g., such as those shown in Fig. 1) are built on the fundamental packet transfer service which is provided by the MAC sublayer, and it is the MAC protocol that determines the characteristics of this fundamental service. Hence, improvements to MAC services result in improved system performance, while the provision of new MAC services means that new applications and services can be developed [10], [11].

A plethora of MAC protocols have been proposed for starcoupled WDM networks (for more details, see [17]). However, most of these MAC protocols are not suited for an integrated services environment because they have been designed with just one *generic* traffic type in mind. As a result, they perform quite well for the traffic streams they have been designed for but poorly for other traffic streams with different characteristics. The purpose of this paper is to combine the advantages of **Delay Requirement**



LOCAL OF LLEDGE

Fig. 1. Multiple traffic stream features and their appropriate MAC protocols.

these various MAC protocols within a single framework so as to be able to efficiently serve a wide range of traffic streams typical in multimedia applications. In particular, we propose a novel WDM MAC protocol that can integrate different types of MAC protocols into a single physical WDM network to efficiently support multimedia traffic streams. This MAC protocol is termed *multimedia wavelength-division multiple-access* (M-WDMA).

The motivation behind integrating different MAC protocols into a single WDM network comes from the fact that different types of traffic streams have different characteristics. As a result, the traffic streams have different transmission and quality of service (QoS) requirements. We can classify multimedia traffic streams as a function of their data burstiness and delay requirements, as shown in Fig. 1. For example, video/audio streams and plain old telephone service (POTS) have small data burstiness but require almost constant transmission delay and almost fixed bandwidth in order to guarantee their QoS. On the other hand, applications such as image networking and distance learning are less stringent in terms of their delay requirements, but their traffic streams are very bursty. Finally, there are other applications that require a very low delay while their traffic streams are bursty. Examples of this type of applications include control messages for video-on-demand systems or interactive games and network control and management.

These different traffic streams are better served by different MAC protocols. Video/audio data streams and other constant-bit-rate (CBR) traffic streams benefit best from prescheduled MAC protocols since they can guarantee that each node has a cyclic and fixed available bandwidth. The best MAC protocols for this purpose would be a simple round-robin time-division-multiplexing access (TDMA) scheme. In a WDM network, a TDMA MAC protocol is applied to each of the channels. The scheduling principle can be represented in a scheduling table that is arranged in such a way that every channel is cyclically allocated to every node in a prefixed order, and every channel is used by a pair of source and destination nodes at a time. This protocol is usually referred to as *interleaved time-division multiple-access* (ITDMA) protocols [3], [4], or WDM/TDM protocols [16]. A typical node configuration for the ITDMA protocols uses a fixed receiver and a tunable transmitter (FR-TT) or a tunable receiver and a fixed transmitter (TR-FT). One drawback of this class of protocols is that if the input load is nonuniform or bursty, the TDMA protocol performs rather poorly.

On the other hand, reservation-based MAC protocols are very well suited for applications where the traffic streams are bursty or the traffic load of the nodes is unbalanced, since reservation-based MAC protocols schedule the transmission according to a particular transmission request. Before a transmission takes place, a node has to send a transmission request to the destination node or to a central controller. Then, the transmission can only start after the source node gets a positive acknowledgment (e.g., grant). Since a lot of detailed information about the transmission can be obtained before the transmission takes place, the transmission schedule can be very efficient and precise. As a result, these MAC protocols can be easily designed to consider tuning time, propagation time, packet processing time, deadline, transmission cost, etc.

The disadvantages of the reservation-based MAC protocols are that the reservation mechanism requires extra bandwidth and time to exchange the request and the acknowledgment information. In addition, the scheduling methods can be relatively complicated, hence, requiring more processing time. These drawbacks cause considerable additional delay and delay variation for packet transmission. Therefore, this type of MAC protocols usually does not perform well for CBR traffic streams or for applications requiring very small delays (for more details about reservation-based protocols, see [13], [16], and [25]).

Finally, random access (contention) MAC protocols usually do not consider the status of the channels or the destination nodes. Once a transmission request emerges, the node starts the transmission almost immediately. This, however, leads to potential collisions between packets. That is, packets being transmitted in the same time slot will collide, and then they have to be retransmitted again. On the other hand, if the transmission succeeds, the packet transmission delay can be extremely low. Thus, random access MAC protocols have the potential of meeting the delay requirements of very urgent messages. Some examples of these urgent messages (e.g., call setup) are listed in Fig. 1. These applications may not generate a lot of traffic data when compared with the other applications. However, once a certain traffic (message) is generated, they require a very low delay. A typical random access MAC protocol that has been proposed in WDM networks is Interleaved Slotted ALOHA (ISA) [4]. These protocols perform very well, especially with respect to packet delays when the traffic load is low. However, if the traffic load gets high, then packet collisions start to occur more frequently. As a result, the performance can be quite poor.

As can be seen, none of these MAC protocols serves all types of traffic well. However, each one of them is ideal for certain



Fig. 2. M-WDMA network architecture.

types of traffic streams. This observation leads us to propose an efficient scheme of integrating all these MAC protocols into a single MAC protocol for WDM networks. We denote this integrated WDM MAC protocol as M-WDMA. In an M-WDMA MAC protocol, an ITDMA-like MAC protocol, a reservationbased (using token passing) MAC protocol, and an ISA protocol are integrated into a single physical WDM network in an efficient way so that different types of traffic streams can use the most appropriate protocol. In our M-WDMA implementation, we consider the following as pects of the environment:

- 1) three types of traffic streams: a CBR, a variable bit rate (VBR) with large burstiness (VBR1), and a VBR with longer interarrival times (VBR2);
- 2) the importance of the deadline associated with each traffic stream;
- 3) a nonzero tuning time of the transmitters/receivers.

In addition, we design a dynamic bandwidth allocation strategy to further improve the utilization of our M-WDMA, which we call M-WDMA+ to distinguish it from the case where the channel bandwidth is allocated in a static fashion.

This paper is organized as follows. Section II introduces our M-WDMA architecture. Section III details the operation of our M-WDMA MAC protocol. In Section IV, we present a simple mathematical model for M-WDMA architecture in order to assess its performance. Section V presents the performance evaluation of M-WDMA networks using extensive computer simulations. Section VI concludes this paper.

II. THE M-WDMA ARCHITECTURE

Consider an M-WDMA network with N nodes which are connected by a star coupler and having C channels, as shown in Fig. 2. Each node has a fixed channel, a home channel, with wavelength $\lambda_i (i = 1, 2, \dots, C)$. The home channels are intended for the destination nodes to receive packets. In case the number of available channels C < N, several nodes ($\lfloor N/C \rfloor$) may share one single home channel. The destination nodes can then accept or discard the packets by checking the addresses associated with these packets.

The M-WDMA MAC protocol has three integrated MAC subprotocols, namely, a TDMA protocol, a reservation-based protocol, and a random-access protocol to better serve all types of traffic streams. In our proposed M-WDMA protocol, each node has three tunable transmitters. The three tunable transmitters are used to serve three different classes of traffic streams according to the corresponding subprotocols. The transmitters are named TDM transmitters (for time-division multiplexing, meaning that these transmitters are driven by a TDM-like protocol), RSV transmitters (for reservation-based protocol), and CNT transmitters (for contention-based protocol).

There are two main reasons why we proposed to use three transmitters and how we specified their operation for our M-WDMA protocol: to reduce penalties of the transmitter tuning time and to combine mulliple subprotocols within M-WDMA. Each one is discussed in detail in the following subsections.

A. Reduce Penalties of the Transmitter Tuning Time

An important parameter that affects the performance of WDM networks is the tuning delay of transmitters. The tuning delay is the amount of time required for a transmitter to tune from one wavelength to another. If we consider fixed-size packets, then we express the tuning delay Γ in units of packet

duration. The value of Γ depends on the transmission rate B (in bits per second), the packet size L (in bits), and the actual tuning delay D (in seconds) through the relationship

$$\Gamma = \frac{DB}{L}.$$
 (1)

Advances in tunable optical devices decreases D, while the trend toward higher rates and smaller packet sizes (e.g., ATM cells) increases B/L. It is difficult to give an exact value of Γ for future networks due to this "push-pull effect of technology." However, if we assume that the proportional decrease of D is proportional to the increase of B/L, then the range of Γ in future networks will be more or less the same as for current networks. As a result, it is safe to predict that the tuning delay of transmitters will also be a problem for future WDM networks. Let us give some example values for D, B, and L. The values that we use hereafter are values frequently used by many researchers in the area [5], [24] and have been used throughout this paper as well: $D = 10 \ \mu s$, B = 1 Gb/s, and L = 424 b (an ATM cell). In this case

$$\Gamma = \frac{DB}{L} = \frac{10^{-5} \times 10^9}{424} \approx 20.$$

Obviously, the value of $\Gamma = 20$ is quite large and can lead to a low throughput and large packet delays in WDM networks if nothing is done to reduce its effect. This same value $\Gamma = 20$ has been used by other researchers when designing their WDM protocols [5], [24]. Most of the existing work to reduce or eliminate the tuning penalty considers a small tuning latency, which is less or equal to 1 (i.e., $\Gamma < 1$) [2], [7], [12], [19]. These solutions do not work well in an environment with large latencies. Some protocols consider arbitrary (small and large) tuning latency [5], [20]. However, they only provide adequate performance for small tuning latencies but not for large tuning latencies, especially when the ratio of the number of stations over the number of channels is not too large. This is not to diminish the novelty and contributions of these protocols as it is impossible to come up with scheduling algorithms that work perfectly well under all conditions (i.e., under any ratio of the number of stations over the number of channels or under any ratio of the tuning time over the packet transmission time).

In order to completely eliminate the tuning penalties, some researchers proposed the use of multiple tunable transmitters at each node [15], [23], [24]. By appropriate scheduling of the tunable transmitters, data packets can be continuously transmitted on all wavelengths. Namely, while one transmitter is transmitting, the others are being tuned to the appropriate channels. This type of transmission is termed *pipelining* [24]. The number of transmitters per node that would eliminate the tuning penalties depends on the ratio of the number of stations over the number of channels and on the ratio of the tuning time over the packet transmission time. As a result, under certain network conditions, to eliminate all the tuning penalties, we are required to use a large number of transmitters per node. In this paper, our approach for reducing or eliminating the tuning penalty follows, in part, the last approach. That is, we proposed to use mul-

tiple transmitters at each node. However, we wanted to limit the number of transmitters per node while eliminating the tuning penalty.

The key question now is what is an appropriate number of transmitters per node that would eliminate the tuning penalty. The authors of [24] have eloquently analyzed the maximum number of transmitters that need to be used per node in order to eliminate the tuning penalty as a function of the ratio of the number of nodes to the number of wavelengths and also the ratio of the tuning time over the packet transmission time.

B. Reduce Penalties of the Transmitter Tuning Time

In our M-WDMA access protocol, we are using three subprotocols (preallocation protocol, reservation protocol, and contention protocol) to access the same optical medium. Each of these MAC protocols accesses the optical medium differently. As a result, they have different threads of control. Our intuition behind this is to allocate one tunable transmitter for each subprotocol in order to simplify the hardware design of the whole interface board (between the WDM network and the terminating equipment) and, in turn, simplify the design of the whole WDM network. As a result, we proposed to use three tunable transmitters per node. Since we are using a constant three transmitters per node independent of the number of nodes, we are not faced with a scalability problem.

Hence, given that we use three tunable transmitters per node (i.e., one transmitter for each subprotocol) and given the operating parameters that we use for D, B, and L, then we have to devise a proper strategy to eliminate the tuning penalties of the transmitters. Our basic strategy to achieve that is to use a frame that consists of multiple number slots. Then, we divide each frame into three segments where each segment is allocated to one of the three subprotocols used in our M-WDMA protocol. In particular, we perform pipelining at the frame level rather than from slot to slot. That is, transmitters perform tuning between frames instead of between slots. With this strategy, channel performance is not affected by the relatively long tuning delays. In fact, the tuning time determines the minimum size of the frame that would allow us to eliminate the tuning penalty (as is shown in Section III-A). In our paper, as we showed previously and as has been used by various other papers [5], [24], we assumed that

$$D = 10 \ \mu s$$
$$B = 1 \ \text{Gb/s}$$
$$L = 424 \ \text{b}.$$

Then, as was calculated previously the value of the transmitter tuning time, $\Gamma = 20$ transmission slots. In order for us to eliminate this 20-slot tuning penalty, using the M-WDMA protocol, under any value of the ratio of the number of nodes to the number of channels, the frame size has to be equal to 30 slots, which is the value that we used in our paper. In this case, while a transmitter is transmitting using its allocated segment, the other two transmitters are tuning to the appropriate wavelength in time to be able to transmit in their allocated segments. However, we must emphasize three points in order to put the paper in a better perspective.

- The main contribution of this paper is the integration of different MAC protocols into a single WDM network and NOT on dealing with the tuning delays. The problem of dealing with tuning delays has been investigated thoroughly by various researchers [5], [24]. The purpose of the integration of different MAC protocols into a single WDM network is to efficiently accommodate various types of traffic streams with different characteristics and QoS demands. This topic, even though extremely important, has received little attention from the research community as far as the design of WDM networks is concerned.
- 2) While we propose to use three tunable transmitters per node under the intuition that it might lead to the simplification (and hopefully lower cost) of the whole interface board (between the WDM network and the terminating equipment), it is difficult for us to quantify that. The main reason for that is the fact that fast tunable transmitters such as the ones envisioned in this paper are purely experimental at this stage. As a result, we cannot accurately estimate their costs, and it would be even more difficult to approximate the portion of costs of an interface card that can be attributed to the tunable transmitters.
- 3) It is very possible to design our M-WDMA protocol using fewer/more than three transmitters per node depending on the given set of parameters (e.g., Γ, D, L, or B). However, doing so will put the paper out of focus. The interested reader can refer to [24] for such research.

In addition to the transceiver configuration, there are many other issues that need to be considered in the M-WDMA architecture design. Because the M-WDMA is a multiple-protocol scheme, using one single queue to store all types of packets causes a severe *head-of-line blocking* problem. To eliminate the problem, we have adopted three types of queues for three differenttypes of traffic. Each type of queue has C queues at each node. As a result, the total number of queues at each node is 3C. To reduce the cost of these multiple queues, they can be easily designed using *random-access memories* (for more details, the reader is referred to [1]). In addition to the C channels, another channel λ_0 is required as a control channel in which networkwide control information can be exchanged. Correspondingly, an extra pair of transmitter-and-receiver at each node is constantly tuned to this channel (the dashed line in Fig. 2). When a packet arrives at a node, it is inserted into the appropriate queue according to its traffic type (e.g., CBR or VBR). When a transmitter is ready, it de-queues the packets from the corresponding queue at a proper time.

In an M-WDMA network, all packets are assumed to have fixed size L b and are transmitted in a networkwide synchronous mode. The transmission time of a packet is referred to as a *time slot*. In other words, the transmission of a packet takes a *slot* time. The slot time is denoted as $T_{\text{slot}} = L/B$, where B is the data transmission rate.

III. THE M-WDMA PROTOCOL AND ITS FRAME STRUCTURE

The M-WDMA MAC protocol is designed to support multiclass traffic streams through the integration of multiple MAC protocols. The implementation issues of the M-WDMA protocol are not simply to have a number of subprotocols within a single network, but also to efficiently utilize the bandwidth by considering the effect of tuning time overhead.

A. The M-WDMA Frame Structure

To complete a packet transmission, wavelength coordination has to be established in a WDM network. In an M-WDMA network, the transmitters have to tune to the home channel of the destination node before the transmission takes place. However, since there is only one receiver at each node, if more than one transmitter sends data to the same receiver, that would produce what is known as receiver collision. When two or more receivers share the same wavelength (the case when N > C), only one of them can be active during the communication;, otherwise, it may cause channel collision. Finally, if at a prescheduled transmission time, the transmitter is occupied by another transmission at the same node, this may result in a *transmitter collision*. Consequently, with the occurrence of any of these collisions, the transmission would be unsuccessful. The major function of the MAC protocol is to properly avoid these collisions and efficiently transmit the packets.

Different types of WDM MAC protocols use different schemes to avoid or resolve these collisions. In a TDMA protocol, packet transmissions avoid collision by properly arranging the schedule maps. In a reservation-based protocol, packet collisions are avoided by actively arranging the packet transmission coordination according to the transmission requests. In a contention-based MAC protocol, when packets collide, they are required to be retransmitted again using, for example, a binary back-off algorithm.

As described in Section II, each node in an M-WDMA network is equipped with three tunable transmitters. Thus, the three transmitters can operate in a *pipeline* fashion. That is, when one transmitter is transmitting a packet, the other transmitters can start their tuning process. As a result, the three types of transmissions (TDM, CNT, and RSV) from a node cannot take place at the same time. This has led us, among other considerations, to organize the three types of transmissions into a time frame. A frame consists of three segments: a TDM segment with length L_{TDM} slots, an RSV segment with length L_{RSV} slots, and a CNT segments with length $L_{\rm CNT}$ slots. The TDM transmissions can only be started in a TDM segment, and analogously for CNT and RSV transmissions, respectively. To make sure that the TDM segments appear in periodic constant intervals, the frame length L_{frame} is fixed (i.e., $L_{\text{frames}} = L_{\text{TDM}} + L_{\text{RSV}} + L_{\text{CNT}}$ is set as a fixed parameter). A transmission example of our M-WDMA MAC protocol is illustrated in Fig. 3.

In this figure, the horizontal direction denotes time and the vertical direction denotes spatial location of the backlogged nodes (the number in the vertical axis represents the location of the nodes, say, 1 means the location of node 1). The white segments denote the TDM segments, the light-shaded segments are the CNT segments, and the dark segments are the RSV



Fig. 3. M-WDMA transmission schedule example.

segments. The numbers in the TDM segments identify the home channel number that the underlying TDM segment ties to, according to the TDM protocol. For example, in the second row and fourth frame, the frame *ID* is 6. It means at that time, the current TDM segment is transmitted on channel 6. Next, in the fifth frame, the frame will be transmitted on channel number 7. All these transmissions happened at node 2, i.e., the TDM transmitter on node 2 is tuned to channel 7 at the time the fifth frame comes. From the figure, we can see, ideally, all channels can be utilized and all nodes can be transmitting even though some of their transmitters may be in a nontransmitting mode (tuning to a certain channel).

The frame length can have an implication on the operation and performance of an M-WDMA network. Hence, it should be properly chosen so that the tuning time can be efficiently masked. That is

$$L_{\text{TDM}} + L_{\text{RSV}} > \Gamma$$
$$L_{\text{TDM}} + L_{\text{CNT}} > \Gamma$$
$$L_{\text{CNT}} + L_{\text{RSV}} > \Gamma.$$

Since $L_{\text{frame}} = L_{\text{TDM}} + L_{\text{RSV}} + L_{\text{CNT}}$, then $L_{\text{frame}} > 3\Gamma/2$. In particular, the length of the frame directly affects the band-

width allocated to TDM segments. Hence, the total TDM bandwidth is given by

$$B_{\rm TDM} = \frac{L_{\rm TDM}B}{(N-1)L_{\rm frame}}$$

When L_{frame} increases, B_{TDM} decreases.

In an M-WDMA MAC protocol, a frame segment is further divided into slots (with length L b) as shown in Fig. 4 so that multiple nodes can share the same segments. Since the TDM

segments and the RSV segments are used in a predetermined manner, then at any one time, only one source node will be using a specific segment, while the CNT segments of a frame can be used by multiple nodes.

The control channel is a shared-access channel that is used by all nodes, which is controlled by a pure TDMA scheme. Its cycle time is T_{ctrl} . A control channel cycle T_{ctrl} consists of N minislots, each of which is assigned to a node. Every node can put on its reserved minislot any necessary information that needs to be known by all other nodes. At the end of a cycle, all nodes receive global information about the status of the whole M-WDMA network. Based on this information, all control procedures can be done by the nodes locally and networkwide synchronously. The control procedures include: 1) reservation requests which are used by the reservation subprotocol; 2) collision acknowledgments; and 3) bandwidth reallocation and notification which is used in an improved version of M-WDMA, named M-WDMA+ (which is discussed in Section III-D). We will discuss the protocol and operation of the control channel in more detail after we depict the three subprotocols.

B. The M-WDMA Protocol

The M-WDMA MAC protocol is an integrated protocol. It includes three subprotocols: A TDM subprotocol, an RSV subprotocol, and a CNT subprotocol. Under the regulation of an M-WDMA frame format, these three subprotocols operate independently.

1) The TDM Subprotocol: The operation of the TDM subprotocol within our M-WDMA network is basically an *interleaved TDMA* MAC protocol [4]. The only differences between our TDM subprotocol and TDMA are that in an M-WDMA network, we take tuning time into consideration,



Fig. 4. M-WDMA frame and slot formats.

and the TDM segment length may change dynamically. Using the M-WDMA protocol, at the border between a TDM segment and an RSV segment, the TDM transmitter starts to tune to the next channel. Since the tuning order is prefixed, no extra information exchange is needed in the control channel. The segment length adjustment is made according to the *dynamic bandwidth allocation algorithm*, which is discussed subsequently.

2) The RSV Subprotocol: In an M-WDMA MAC protocol, the RSV packet transmission is controlled using a *multiple token* method [25]. Each channel is associated with a "token." A node can send its packets onto the destination channel only if it holds the corresponding token. An obvious advantage of using a token-based scheme in our RSV subprotocol is that it can efficiently support very bursty traffic streams, and its implementation can be simple. The disadvantage of this scheme is that the channel efficiency may not be high when compared with a perfect scheduling scheme, since in a frame, only one single node is allowed to access the RSV segment at a time.

In our implementation of the multiple-token mechanism, the tokens are not explicitly issued. They are in fact *virtual* tokens. In each cycle of the control channel, the M-WDMA nodes broadcast their transmission requests to all nodes. At the end of the cycle, all nodes synchronously execute a *token rotation algorithm* to determine the token distribution in the coming frame. The token rotation algorithm is depicted in Fig. 5.

Here, a prefixed order can adopt any order. However, we usually use an order of node *ID* so that all nodes can obtain identical results when the algorithm is executed in a distributed manner. If an M-WDMA network has an unbalanced load (e.g., client–server traffic), we can change the token order to favor the loaded nodes. In this way, more transmission chances are allocated to those nodes. It is worth noting that there are no transmission conflicts among nodes when using the token rotation algorithm since each node can only send one transmission request, and there is only one token available for any particular channel. The last part of the token rotation algorithm is to arbitrarily allocate the spare token to some idle nodes. To ensure that [Token Rotation Algorithm]

Put all the tokens with their holder into a token list;

Put all the nodes with a request into checking list;

For every token do {

Check if the token holder still needs the token

If (it is true and does not exceed the token holding time)

then {

allocate the token to the current holder;

remove the node from the checking list;

remove the token from the token list; continue to the next token;

};

Find another node waiting for the token in a pre-fixed order;

if (Found) {

Allocate the token to that node.

remove the node from the checking list;

remove the token from the token list;
}

For each of the token remaining in the token list, do {

Find a node in the checking list, which has no token;

Allocate the token to the node;

remove the node from the checking list; remove the token from the token list; }

Fig. 5. Token rotation algorithm.

the algorithm gets identical results, the allocation is also carried out in a prefixed order.

The token rotation is executed according to the control channel timing. Thus, the rotation results have to be synchronized with the data transmission (i.e., the frame timing). Hence, any rotation results are effective for the first coming RSV segment after each execution of the token rotation algorithm.

3) The CNT Subprotocol: The CNT subprotocol of our M-WDMA MAC protocol is similar to the interleaved slotted ALOHA [4]. The active nodes compete for the slots in the current CNT segment. In case there is a collision, retransmission is scheduled according to a binary back-off algorithm. In the context of the M-WDMA, two problems have to be solved. The first one is how to handle the acknowledgments. In M-WDMA, there are two receivers at each node: one is the home channel receiver, and the other is the control channel receiver. Although we can detect the collision at the receiver side, the notification of collision can consume a portion of the control channel bandwidth. In our design, the collision notification is issued in the early part of the control channel cycle. Considering the round-trip propagation delay, the total acknowledgment delay is the propagation time plus the control channel cycle time. In case the propagation delay is large, we can adopt a larger sliding window scheme to reduce its affect.

The second problem that needs to be addressed is the retransmission of collided packets. When a collision occurs, retransmission is inevitable. However, the scheduled retransmission should not cross the border of a CNT segment; otherwise, it will collide with other types of transmissions. However, if the back-off is limited to within one CNT segment, it may cause even more collisions. Thus, a CNT segment counter is set at each node to resolve this problem. When a node encounters the last slot of a CNT segment, its CNT segment arrives, the counter starts to tick again. In this way, the retransmission time is calculated in terms of the counter rather than the real slot numbers in a frame. As a result, retransmission can be carried out across the frames.

C. Control Channel Configuration

The control channel operates in a TDMA manner independently from the data transmission. One cycle consists of N minislots, each of which is designated to a node. The cycle length is determined by the amount of information required and the number of nodes involved. Since the TDM subprotocol is a preallocated protocol, there is no need to exchange control information before transmission through the control channel. The RSV subprotocol is a multiple-token passing protocol, so the information needed to process the token rotation is only the states of the queue. We can use a bit map to represent the reservation request, each node taking a bit. If the bit is "1," that means that the queue is nonempty, and there is at least one packet waiting for transmission; otherwise, when it is "0," that implies the queue is empty and there are no transmission requests. Therefore for each node, one bit is sufficient. If there are N nodes in the network, each minislot of the control channel needs N b to indicate the requests information (including itself for simplicity). For the CNT transmission acknowledgment, we can use the bit map, again, to indicate the success or failure during the contention. Since there are $L_{\rm cnt}$ slots in a frame for contention, then $L_{\rm cnt}$ bits are enough for a CNT transmission acknowledgment. If a bit is "1," it implies the corresponding slot is successful. Since this information is broadcast to all nodes through the control channel, the source node can obtain this information and start to prepare the next transmission (or tuning). If a bit is "0," then there is a collision in the corresponding slot; thus, the source nodes involved in this collision have to start the retransmission process. Consequently, the length of a minislot is determined by $N + L_{\rm cnt}$. As a result,

$$T_{\rm ctrl} = \frac{N(N + L_{\rm cnt})}{R}$$

where B is the channel bit rate. To realize the frame-by-frame reservation and collision detection, the control cycle has to complete within a frame time, that is,

$$\frac{N(N+L_{\rm cnt})}{l_{\rm slot}} < L_{\rm frame}$$

where L_{slot} is the slot length in bits. Solving the inequality, we get

$$N < \frac{L_{\rm cnt}}{2} \left(\sqrt{\left(1 + \frac{4l_{\rm slot}L_{\rm frame}}{L_{\rm cnt}^2} \right)} - 1 \right).$$

This leads to a network scale limitation. For example, if we choose, $L_{\rm cnt} = 10$ slots, $L_{\rm frame} = 30$ slots, $l_{\rm slot} = 424$ b, then the number of nodes should be N < 107.

By adding another receiver to do the collision detection, the $L_{\rm cnt}$ can be eliminated, then the scale limitation becomes $(l_{\rm slot}L_{\rm frame})^{1/2}$ (which is 11, as in the previous example). Alternatively, by releasing the requirement of frame-by-frame reservation, the square-root item can increase considerably, as a result, the restriction becomes more relaxed.

D. Dynamic Bandwidth Allocation: M-WDMA+

Due to the randomness of the traffic streams, the bandwidth required by a particular traffic stream cannot be accurately estimated. It is possible that some of the segments are overloaded, while some segments are partially filled, resulting in a low efficiency. Although some of the prediction methods based on long-term statistics [14] are useful in this context, dynamic and adaptive solutions seem to be more appropriate, since they can match the transmission requirements more accurately. In an M-WDMA network, the frame length is fixed, which implies that utilizing all the frame slots must result in the full utilization of all available bandwidth. Therefore, for a given set of traffic streams, if we can properly allocate the portion of bandwidth that the three types of segments should take, then all available bandwidth would be properly utilized. Fortunately, in an M-WDMA, different classes of transmissions are grouped into a single frame, so the adjustment can be easily realized by adjusting the segment sizes.

In order to accomplish this dynamic adjustment, we have to be able to detect the traffic load. That is, we should be able to determine the segment size required by the traffic streams so [Dynamic Bandwidth Allocation Algorithm] Given $\stackrel{\wedge}{l_{\text{TDM}}}$, the number of TDM requests $\stackrel{\wedge}{l_{\text{RSV}}}$, the number of RSV requests $l = L_{\text{frame}}$ 1. if $\stackrel{\wedge}{l_{\text{TDM}}} < Q_{\text{TDM}}$ then $L_{\text{TDM}} = \stackrel{\wedge}{l_{\text{TDM}}}$; else $L_{\text{TDM}} = Q_{\text{TDM}}$; $l = l - L_{\text{TDM}}$; 2. if $\stackrel{\wedge}{l_{\text{RSV}}} < Q_{\text{RSV}}$ then $L_{\text{RSV}} =_{\text{RSV}}$; else $L_{\text{RSV}} = Q_{\text{RSV}}$; $l = l - L_{\text{RSV}}$; 3. $L_{\text{CNT}} = l$;

Fig. 6. Dynamic bandwidth allocation algorithm.

as to meet their QoS requirements. By looking into the characteristics of the different types of the subprotocols within an M-WDMA network, we can find the solution. The CBR traffic is connection oriented in nature. Hence, once a CBR connection is established, the connection is usually kept for a relatively long time. During the connection time, the bandwidth required for the connection is known. Then, the number of slots required in each frame can be derived before the transmission takes place. For the VBR traffic streams using the reservation-based subprotocol, the required bandwidth varies a lot because of the burstiness of the data. However, the reservation subprotocol requires the source node to submit its transmission request before the transmission can proceed. In an M-WDMA network, the transmission requests are broadcast to all other nodes through the control channel. Therefore, every node can obtain the number of slots required for the RSV segments in every frame, before the transmission is started. Having determined the TDM and the RSV segment lengths, the remaining slots of the frame are left to the CNT segment. As a result, before a frame is used, we already know the sizes of all its segments.

Once the segment sizes required are known, bandwidth allocation becomes straightforward and more flexible. Our strategy is to assign priorities to the subprotocol segments in the following order: the TDM segments, the RSV segments, and then the CNT segments. That is, the TDM requirements are considered first (we allocate CBR traffic first). Next, we serve the VBR traffic streams through the RSV segment if there is bandwidth left. Finally, the remaining bandwidth, if available, is all given to the CNT segment. This allocation algorithm is described in Fig. 6.

In this algorithm, Q_{TDM} and Q_{RSV} are system parameters that indicate the maximum number of slots that a TDM segment and a RSV segment can have. These two parameters reflect the bandwidth allocation strategy. Generally, these two parameters should satisfy

$$Q_{\text{TDM}} > 0$$

 $Q_{\text{RSV}} > 0$
 $L_{\text{frame}} - Q_{\text{TDM}} > \Gamma.$

Fig. 7 shows one example of applying our dynamic bandwidth allocation algorithm. The histogram is obtained using the following parameters: $L_{\text{frame}} = 30$, $Q_{\text{TDM}} = 14$, $Q_{\text{RSV}} = 25$,

 $\Gamma = 16$, and the normalized traffic load is 30%. From the histogram, we can see that more bursty data can be efficiently accommodated and more available bandwidth can be exploited by the CNT subprotocol when compared with the static allocation scheme (i.e., M-WDMA). The subprotocol with the highest priority, the TDM subprotocol, just uses the bandwidth it needs.

To determine the segment configuration for each frame in a distributed manner, every node has to record the current traffic load in each frame and in each channel. In this case, the information required for reservation in the RSV segment is not only the transmission of the request, but also the number of slots requested. This results in a longer control cycle and longer transmission delay. To solve the problem, we split the request part into two alternating stages. In first stage, the minislot is the same as in M-WDMA where a minislot consists of a request bit map and a CNT acknowledgment bit map. At the end of this cycle, the token rotation is determined. Then, the RSV transmitter can start to tune to the destination channel. In the next cycle, the second stage, the minislot consists of *number of requests* and the CNT acknowledgment bit map. Because the number of slots requested is needed only at the beginning of the frame in which these requested slots are loaded, and the second stage is executed in parallel with the transmitter tuning since the end of first stage, the necessary information for the RSV segment can all be available on time without extra delay due to the extra stage in the control channel. The number of requested slots is expressed in $l_{\rm req}$ bits. In the worst case, where all tokens are requested, the total number of nodes that can be supported is $1/l_{req}$ of that in M-WDMA. Hence, there is a tradeoff between the network scalability and utilization of the network. If the l_{req} bits takes l slots (l > 1) as the unit, then for the same range of burst length, fewer bits are needed, but the utilization would be degraded because the allocation is carried out in the units of l, no matter whether it is necessary or not. On the other hand, if l = 1, no wastage is incurred but longer l_{req} bits are required for the same range of bursty length. In the remainder of the paper, we will eliminate the boundary effects and assume the control cycle is sufficient for a given network size with frame-by-frame dynamic allocation.

IV. MATHEMATICAL MODELING OF THE M-WDMA NETWORK

This section investigates the performance of the M-WDMA MAC protocol through simple mathematical models. We focus our attention on determining the packet transmission delays using our M-WDMA MAC protocol. The dynamic allocation of bandwidth is not considered here. This is left for a sequel of this paper. In an M-WDMA MAC protocol, three subprotocols coexist together, but they are not active simultaneously. When one of the subprotocols is in use, the other subprotocols are idle from the point of view of the channels and the nodes. Given that the bandwidth (subprotocol segment sizes) is statically allocated in the M-WDMA, the three subprotocols can be thought of as if they operate using three different networks. These three *virtual* networks have a bandwidth equal to the bandwidth portion that is allocated to that subprotocol in an M-WDMA network. This is reasonable under the assumption that the three classes of traffic streams are independent from



Fig. 7. Histogram of an M-WDMA+ bandwidth occupation.

each other. We term this assumption as *protocol independence assumption*. With this assumption, the three subprotocols can be modeled separately.

A packet has to wait in its queue until it is its turn to be served, no matter which subprotocol would serve it. That is, the server is not always available. It simply alternates between the states of *idle* and *busy*. In other words, a server (transmitter) would poll the queues containing the packets each time it can serve a packet. As a result, it is reasonable to apply a polling system analysis in our case [8], [9], [21], [22] for modeling our M-WDMA network. We assume that all the nodes in the M-WDMA network have equal probability to generate packets and also have equal probability to be the destination of a packet. Let λ be the network normalized traffic load, and λ_{TDM} , λ_{RSV} , and λ_{CNT} be the mean traffic loads for the individual segments of the subprotocols, respectively. The relationships between these values are

$$\begin{split} \lambda_{\text{TDM}} = & \frac{L_{\text{TDM}}\lambda}{L_{\text{frame}}} \\ \lambda_{\text{RSV}} = & \frac{L_{\text{RSV}}\lambda}{L_{\text{frame}}} \\ \lambda_{\text{CNT}} = & \frac{L_{\text{CNT}}\lambda}{L_{\text{frame}}}. \end{split}$$

Without loss of generality, we use λ_i to denote the mean traffic load that arrives into queue *i*. The service time for queue *i* is called the *visit time* of queue *i*, a generally distributed random variable B_i with mean b_i and second moment $b_i^{(2)}$. Since the server visits the queues in a cyclic order, the reference to the queue indexes is done *modulo* N. After visiting queue i, the server switches to queue i + 1. The period during which the server switches from queue i to queue i + 1 is called the switch-over period, denoted by a random variable S_i , which is assumed to be a generally distributed random variable with a mean s_i and a second moment $s_i^{(2)}$. It is assumed that the packet arrival times, the packet service times, and the switch-over processes are mutually independent. For a particular queue i, the period between two consecutive visits to queue *i* is named *i cycle*. The duration of an *i cycle* is denoted by Y_i . Let R_i be the residual time in an *i cycle*. Our purpose here is to get the mean packet delay $E[W_i]$ in queue *i* in steady state. Let X_i denote the number of packets present in queue *i* at an arbitrary moment. Under the assumption of exhaustive service, the waiting time of an arbitrary packet at queue *i* consists of the following two components:

- the time since the packet arrives until the next visit of the server to queue *i*, that is, *R_i*;
- 2) the time required for the server to serve the X_i packets found at the queue during the last visit time.

Fig. 8 illustrates the relationship between these time components.



Fig. 8. Transmission delay components of a tagged packet.

By observation, we have

$$E[W_i] = E[R_i] + E[X_i].$$
⁽²⁾

By Little's law, we get

$$E[X_i] = \lambda E[W_i]. \tag{3}$$

This is true according to the Poisson arrivals see time averages (PASTA) [9] property. When an arbitrary packet arrives at an arbitrary time, the number of packets ahead of it in the queue is also considered the mean number of packets in the queue.

We approximate $E[R_i]$ as [9]

$$E[R_i] = \frac{E[Y_1]E[R_1]}{E[Y_i]}.$$
(4)

Consequently, we can obtain a general form for the mean packet delay which is given by

$$E[W_i] = \frac{E[R_i]}{(1 - \lambda_i)}.$$
(5)

Next, we apply this general form onto the individual subprotocols of the M-WDMA network to finalize our mathematical model.

A. The TDM Subprotocol Model

In the TDM subprotocol, the transmission is organized according to a pre-fixed schedule table, as shown in Fig. 3. In addition, the cycle time is fixed to $(N - 1)T_{\text{frame}}$. Since the schedule for transmission operates in a round-robin scheme, each queue has a fair chance (i.e., preallocated bandwidth) to transmit packets and the TDM segment is of fixed length L_{TDM} . Hence, when an arbitrary packet arrives at an arbitrary time, the expected residual time is given by

$$E[R_i] = \frac{(N-2)T_{\text{frame}}}{2}.$$
(6)

Note that the residual time excludes a frame time that is occupied by the underlying node. Using (5), we can get the mean packet delay using the TDM subprotocol as

$$E_{\text{TDM}}[W_i] = \frac{(N-2)T_{\text{frame}}}{2(1-\lambda_i)}.$$
(7)

B. The RSV Subprotocol Model

Using the RSV subprotocol of the M-WDMA network, the *i* cycle is no longer fixed. It depends on the elapsed time the token stays on the other nodes under the control of the RSV subprotocol. When a node gets a token, it may occupy all the slots of the RSV segments in the current frame and may continue to use the RSV segment in the next frames. In case few or no other nodes use this channel, the token may come back again. Hence, the *i* cycle could have large variations. According to the suggestion given in [9], the $E[Y_i]$ can be derived using the recursive formula

$$E[Y_i] = b_i + \sum_{j \neq i} \min(1, \lambda E[Y_j])b_j + s \tag{8}$$

where s is the overall switch-over time. Here, we assume s = 0, since when L_{frame} is used to calculate the cycle, the tuning time (switch-over time) has already been included.

Since we have a fixed frame length, $E[R_1]$ can be chosen in the same way as in the previous subsection. Combining $E[R_i]$ and $E[Y_i]$ into (5), we can get the mean delay of packets $E_{\text{RSV}}[W_i]$.

C. The CNT Subprotocol Model

The CNT subprotocol in an M-WDMA network is a random access protocol. Every node can access the CNT segment of each frame, so the *i cycle* time is T_{frame} . Due to collisions, however, packets' retransmissions are needed. As a result, the network load is no longer equal to just λ_{CNT} . Many mathematical models have been proposed for such a protocol [4]. In our case, we reasonably follow Dowd's model, which is a *semi-Markov* process model. According to [4], the average number of packets in the system is given by

$$E(N) = \sum_{i=0}^{2(B+1)} E[N_i]P_i$$

where P_i is the probability of being in state S_i , where the notation of system state S_i implies that there are *i* packets residing in the buffer. *B* is the capacity of the buffer. By Little's law, the mean delay is given by

$$E[X_i] = \frac{E[N]}{\Gamma_s} \tag{9}$$

where $\Gamma = \gamma \sum_{i=1}^{B+1} P_i$, and γ is the probability of successful transmission.

If we substitute (9) into (2) and given $E[R_i] = T_{\text{frame}}/2$, we can obtain the mean packet delay using the CNT subprotocol of our M-WDMA network.

D. Maximum Throughput of the Subprotocols

Since the TDM and the RSV subprotocols are collision-free MAC protocols and all the slots in their corresponding segments could be fully utilized, the maximum throughput should be

$$S_{\rm TDM,max} = \frac{CL_{\rm TDM}}{L_{\rm frame}}$$
(10)

$$S_{\rm RSV,max} = \frac{CL_{\rm RSV}}{L_{\rm frame}}.$$
 (11)

Using the CNT subprotocol, we can have packet collisions, and packet retransmission would be needed. Thus, the maximum throughput cannot exceed 1/e = 36.8% [4]. As a result,

$$S_{\rm CNT,max} = \frac{0.368CL_{\rm CNT}}{L_{\rm frame}}.$$
 (12)

V. PERFORMANCE EVALUATION OF M-WDMA THROUGH SIMULATION

In this section, we evaluate the performance of our M-WDMA using extensive discrete-event simulations. The purpose of the simulations includes the validation of the mathematical models discussed in the previous section, the quantitative evaluation of the improvement gained by our M-WDMA+ when compared with M-WDMA, and the comparison of the performance of M-WDMA+ with conventional WDMA MAC protocols.

A. Network Parameters

Table I lists the M-WDMA network parameters that have been used in our simulation.

Note that in the table, the frame time is $12.72 \ \mu s$. This implies that if a connection can get one slot in each frame, the bandwidth obtained would be 33.3 Mb/s. For guaranteed bandwidth allocation (TDMA), if the number of nodes is N = 10, and C = N, the rate of one slot per frame corresponds to a 3.3-Mb/s data rate (ten frames per cycle). This rate can already support two MPEG-I data streams (each around 1.6 Mb/s). Similarly, for an 8-KHz phone call, a one-slot-per-frame rate means that we can support about 200 pairs of phone conversations simultaneously. Higher bandwidth can be obtained in multiples of 3.3 Mb/s by allocating one more slot per cycle.

The peak rate can be obtained through the allocation of slots using the RSV segment. In case the RSV segment size is ten slots, the peak rate can reach 333 Mb/s. Even larger peak rates can be obtained by further extending the RSV segment. In an M-WDMA network, the theoretical available peak rate is 1 Gb/s. In such a case, the whole frame becomes the RSV segment.

TABLE I M-WDMA NETWORK PARAMETERS USED IN OUR SIMULATION

Parameters	Value
Channel data rate	1 Gb/s
Slot size	53 Bytes
Frame size	30 slots
Slot time	424 ns
Frame time	12.72 μs
Number of Channels	С
Tuning Time	20 slot time
Number of nodes	Ν
Maximum buffer size	200 slots

As mentioned in the introduction of this paper, urgent messages requiring very small delay would be sent through the CNT segment where a random access protocol is applied. In case no collision occurs, the response time can be expected to be equal to the round-trip propagation delay. Theoretically, when the traffic load for the slotted ALOHA is less than about 20% of the total CNT bandwidth, the collision rate can be very small. As the contention slots can be used at any time, the available effective throughput in the case of one slot per frame can be expected to reach about 7 Mb/s. It is worthwhile to mention again that the benefit from the CNT access is for a very small delay and simple access control, not high throughput.

B. Traffic Generation Models

In our simulation study, three types of traffic streams are generated: the CBR sources, the VBR sources with small burstiness (VBR1), and the VBR with large burstiness (VBR2). For the CBR packet sources, the packets are generated periodically. The period is determined according to the traffic load percentage which corresponds to the system parameter λ . The sources and destinations are randomly chosen in a uniform distribution. The unit of the traffic generation is slot time.

The VBR1 data sources are in the class of connectionless traffic streams with small amounts of data, but expecting quick response, e.g., game control signaling and network management and signaling. The traffic load from these interactive applications is light but may be bursty. In our simulation, the generation of VBR1 traffic follows a Poisson arrival process. In addition, the packets are generated at each node in a batch as a message. The length of a message is short.

The VBR2 data are generated using an ON–OFF state machine with a Poisson arrival process and a uniformly distributed random ON state duration. The burstiness length of the traffic is a system parameter. This is used to emulate heavy data transfers with less time restrictions, such as data or image transfer and offline video/audio transmission.

C. Model Validation

Fig. 9(a) and (b) shows the mean delay and throughput of an M-WDMA network. These results are obtained using our mathematical model and using simulation. For simplicity, in both the simulation and mathematical model, we assume the propagation delay to be zero. The number of nodes is N = 10 and the



Fig. 9. Performance of the M-WDMA protocol with the fixed bandwidth allocation. (a) Mean delay. (b) Throughput.

number of channels C = 10. We also assume that the frame size is $L_{\text{frame}} = 30$, and the subprotocol segment sizes are $L_{\text{TDM}} = 10$, $L_{\text{RSV}} = 10$, and $L_{\text{CNT}} = 10$.

As can be seen from the figures, our simulation and mathematical model results are close to each other. This is an indication of the accuracy of our simple mathematical models for the M-WDMA MAC protocol.

D. Performance Evaluation

1) Improvements With M-WDMA+: By dynamically allocating the M-WDMA bandwidth to different types of traffic streams, the network efficiency and performance can be improved as discussed in Section III. The numerical simulation quantifies this improvement as illustrated by Fig. 10(a) and (b).



Fig. 10. Performance improvement of M-WDMA+ over M-WDMA. (a) Mean delay. (b) Throughput.

In our simulation, the three different types of traffic streams are generated: CBR, VBR data streams (VBR1), and lowarrival-rate traffic data (VBR2). The CBR traffic is controlled by varying the stream length and the mean traffic load. The VBR1 traffic is controlled by varying the burstiness of the data (e.g., from 1–20 slots). The VBR2 traffic is similar to the VBR1, but its mean load is smaller. More precisely, the three types of traffic streams are such that 47% of the network load is the CBR data, 47% is VBR1 data, and 6% is VBR2 data. As a result, the bandwidth allocation in the M-WDMA also takes exactly the same proportion for the three subprotocols, respectively.

By applying the dynamic bandwidth allocation algorithm, the channel bandwidth is utilized more efficiently. Hence, the performance can be improved tremendously. From these figures, we can see that the performance of the CBR streams using the M-WDMA and the M-WDMA+ MAC protocols are more or less the same since both of them guarantee the bandwidth required for the CBR traffic streams. However, the performance of the VBR1 traffic streams is improved since the M-WDMA+ can accommodate longer data burstiness than in the "static" M-WDMA. The most significant improvement is with regard to the VBR2 traffic streams, where a random access subprotocol is applied. This is because all the unused bandwidth can be fully utilized by the VBR2 traffic data. For a given traffic load, a larger bandwidth implies lower collision rates and thus better efficiency.

2) Performance Comparison of M-WDMA+ With ITDMA: In this subsection and the following ones, we want to illustrate the superiority of integrating various protocols into a single one (M-WDMA) when compared with any individual conventional protocol for a WDM network. We now compare the performance of the M-WDMA+ with the ITDMA protocol. The ITDMA protocol can be found in [4], and it is exactly the same as the TDM subprotocol in our M-WDMA+ network. In other words, the ITDMA can be looked at as our M-WDMA, where all the frame size is a TDM segment. For a fair comparison, exactly the same traffic patterns and proportions are loaded onto both networks, and the same node configurations are also applied, i.e., in ITDMA, the transmitters are also pipelined. The mean delay and loss rate for both networks are shown in Fig. 11(a) and (b).

The CBR traffic performance is more or less the same for both networks, since our M-WDMA+ MAC protocol takes advantage of the ITDMA MAC protocol for this type of traffic. But for other types of traffic streams, the VBR1 and the VBR2, they are much better served by our M-WDMA+ network. In Fig. 11(a), we can see the three different traffic schemes obtain very similar performance using the ITDMA, i.e., the delay is almost constant. This is not good because some urgent messages may unnecessarily wait too long. On the other hand, the M-WDMA+ can better adapt to the traffic characteristics. You may see that when the traffic load is not very high, the delay can be quite low for the RSV and the CNT transmissions. Correspondingly, the advantage is also obvious in the loss rate comparison, as shown in Fig. 11(b). In the high traffic load, we can observe that the performance of the RSV traffic of the M-WDMA+ is a bit worse than that of the ITDMA. This is because the ITDMA does not need to do reservation. While for the TDM traffic, the M-WDMA+ is better than the ITDMA because the head-of-line problem is more severe in the ITDMA. The most significant advantage of the M-WDMA+ over the ITDMA protocol is the CNT part because the M-WDMA+ supplies much more bandwidth for the traffic contention, thus reducing collision.

3) Performance Comparison of M-WDMA+ and MT-WDMAC: We now compare the performance of our M-WDMA+ protocol with a reservation-based MAC protocol, namely, MT-WDMA [25]. The MT-WDMA stands for multiple-token WDMA. This protocol has been described previously and is similar to the one we use as part of our integrated

MAC protocol for the M-WDMA. The only difference is that all the traffic streams are handled by a unique MT-WDMA protocol in this case, as opposed to multiple subprotocols for the case of the M-WDMA.

By examining Fig. 12(a) and (b), it is obvious that the M-WDMA+ performs better for all types of traffic streams because the MT-WDMA protocol has extra time overhead for the reservation. The RSV part in the M-WDMA+ is better than in the MT-WDMA, since the RSV protocol in the M-WDMA+ is less affected by the head-of-line blocking problem. The TDM part is better because the CBR traffic in the M-WDMA+ avoids the reservation operation. Comparing the CNT parts using the MT-WDMA and the M-WDMA+, we can see that the M-WDMA+ supplies much better performance because the M-WDMA+ gives more bandwidth for packet contention. Again, this is an affirmation of the superiority of using an integrated approach when designing MAC protocols for multimedia applications.

4) Performance Comparison of M-WDMA+ With ISA: Here, we compare the integrated MAC approach for our M-WDMA+ with that of a single random access protocol, ISA [4].

ISA is implemented using the same network configurations as our M-WDMA+ protocol. Again, the difference is that in ISA, all types of traffic streams are processed under the same MAC protocol, unlike the M-WDMA+ MAC protocol, where each type of traffic stream is allocated to the proper subprotocol. The results shown in Fig. 13(a) and (b) clearly illustrate the advantages of our integrated MAC approach in M-WDMA+. Note that only when the traffic load is very light, the ISA can achieve lower delay and similar loss rate as the TDM part in M-WDMA+.

5) Performance Comparison of M-WDMA+ With Conrad: CONRAD is a state-of-the-art MAC protocol that is proposed for a passive-star-based single-hop WDM local lightwave network [26]. It can efficiently provide services to support data traffic with tight delay constraints, i.e., real-time traffic, directly in the optical layer. In the original paper of CONRAD protocol, only two types of traffic are generated to evaluate its performance. In order to compare its performance with that of M-WDMA+ protocol, we have included three types of traffic, i.e., CBR traffic, VBR1 traffic, and VBR2 traffic. Under the same environment with the same system parameters, we compare the performance of both protocols.

The parameters of the simulation experiments for comparison purposes are as follows. The number of nodes in the system is ten. The number of channels is also ten. The time unit in the simulation is the same as the time slot for the transmission time of one packet. Propagation time is assumed to be zero in order to compare the silent feature of both protocols. The transceiver tuning time is 16 time units. CBR traffic, VBR1 traffic, and VBR2 traffic are generated for both protocols. We consider CBR traffic as real-time traffic, which occupies 30% of total traffic. The VBR1 traffic is 17% in total. The remainder is the VBR2 traffic. For the CBR traffic stream, it is assumed that a message, which consists of several packets, is generated periodically. The message is always considered with ten packets. VBR1 and VBR2 traffics are generated as Poisson processes.



Fig. 11. Performance comparison between M-WDMA+ and ITDMA. (a) Mean delay. (b) Loss rate.

The length of a message from VBR1 traffic follows an exponential distribution with mean as five packets. The length of a VBR2 message also follows an exponential distribution with mean as 15 packets. We assume that the destination nodes are randomly chosen among ten nodes following a uniform distribution. For a fair comparison, exactly the same traffic pattern and proportions are designed for both protocols. We present the simulation results in terms of network throughput and average packet delay of two protocols in Fig. 14.

Fig. 14(a) shows the average packet delay versus normalized offered load for both protocols. An interesting observation from the figure is that by using the CONRAD protocol, the delays of VBR1 and VBR2 are almost convergent together. This is because the packets from VBR1 and VBR2 streams enter into



Mean Delay Comparison: RSV vs M-WDMA+

Fig. 12. Performance comparison between M-WDMA+ and MT-WDMA. (a) Mean delay. (b) Loss rate.

one queue whose discipline is first-in-first-out. We also observe that using the CONRAD protocol, only with very light traffic, the average packet delays for VBR1 and VBR2 traffics are small. The average packet delay for CBR traffic is smaller than those of VBR1 and VBR2 traffics. We can also find that the average packet delays for the three types of traffics using the M-WDMA+ protocol are always smaller than those using the CONRAD protocol. Using the M-WDMA+ protocol when the traffic load is moderate (the normalized offered load is less than 0.6), the average packet delays are not seriously affected when there is a small increment of offered traffic. The delay of CBR traffic is larger than the delays of VBR1 and VBR2. The delay of VBR1 traffic is the smallest among three types of traffic. However, when the traffic load is heavy, the delay of VBR1 traffic is increased significantly and it becomes larger than those of other two traffics. The reason is that the VBR1 traffic uses the CNT subprotocol, which is designed based on a slotted ALOHA scheme. It is well known that under light traffic conditions, the delay in slotted ALOHA could be small. Under heavy traffic load, many more collisions would occur, making the delay large. In summary, the overall performance of the M-WDMA+ protocol is always better than that of CONRAD protocol in terms



Fig. 13. Performance comparison between M-WDMA+ and ISA. (a) Mean delay. (b) Loss rate.

of average packet delay for all three types of traffic in the network. The primary reason for it is that the M-WDMA+ protocol has employed three subprotocols to adapt to different types of traffic and three transceivers to cover the overhead of the tuning time of them.

From Fig. 14(b), we can see that the M-WDMA+ protocol always has higher throughput than that of the CONRAD protocol. Using the CONRAD protocol, the overall throughput reaches the maximum value when the normalized offered load is about 0.2. Using the M-WDMA+ protocol, the overall throughput keeps increasing and reaches the maximum throughput as the traffic load approaches 1.0. The reason of these facts is that by using the CONRAD protocol, the transceiver has to take time to tune to the specified channel for data transmission. This action makes transmission channels idle for some time slots. It results in low throughput, even under heavy traffic. It is obvious that when the normalized offered load is more than 0.2, the throughput is not increasing any more. However,



Fig. 14. Performance comparison between M-WDMA+ and CONRAD. (a) Mean delay. (b) Throughput.

there is no such thing when using the M-WDMA+ protocol, simply because of the pipeline working style of the group of transceivers.

6) Performance of M-WDMA+ Under Unbalanced Traffic: In the previously described simulation experiments, we evaluated the performance of the M-WDMA+ protocol under the assumption that the network is symmetrically loaded. The traffic ratio of three types of traffic is always the same as TDM : RSV : CNT = 7 : 7 : 1 to every node. In this subsection, we present one group of results from the simulation experiment, in which unbalanced traffic is applied to the network. We consider that the traffic ratio of the three types of traffic to be different to different nodes. We compare the performance of our proposed protocol under the balanced



Mean Delay Comparison: Balanced vs Unbalanced Traffic

Fig. 15. Performance comparison between balanced and unbalanced traffic for M-WDMA+. (a) Mean delay. (b) Throughput.

and unbalanced traffic to show that the performance of the M-WDMA+ protocol can still be good enough to support multimedia traffic even under the unbalanced traffic load.

In this simulation experiment, we assume that in the network, the traffic ratio for three nodes is TDM : RSV : CNT = 14 : 14 : 2, the traffic ratio for another three nodes is TDM : RSV : CNT = 13 : 13 : 4, and the traffic ratio for thr other four nodes

is TDM : RSV : CNT = 15 : 14 : 1. The simulation results have been shown in Fig. 15 as follows.

From Fig. 15(a), we can find that under unbalanced traffic condition, for CBR and VBR2 traffic, the average packet delays are quite lower than those under balanced traffic. However, the average packet delay for VBR1 traffic is quite higher than that under balanced traffic. The reason for these facts is obvious.

Under the unbalanced traffic condition with the ratio of three traffic loads as TDM : RSV : CNT = 14 : 14 : 2 for three nodes, TDM : RSV : CNT = 13 : 13 : 4 for three nodes, and TDM : RSV : CNT = 15 : 14 : 1 for four nodes, the amount of the VBR1 traffic is more than that under the balanced traffic condition with the traffic ratio as TDM : RSV : CNT = 7: 7 : 1 for the nodes. The M-WDMA+ protocol has allocated more bandwidth to the VBR1 traffic. However, due to the feature of CNT subprotocol, the VBR1 traffic can only experience lower delay when the traffic load is lower than 0.6. When the traffic load gets higher than 0.6, the delay of the VBR1 traffic experiences higher delay. On the contrary, the network has allocated less bandwidth to VBR2 traffic and CBR traffic because the amounts of the VBR2 traffic and CBR traffic under the unbalanced traffic condition are less than those under the balanced traffic condition. It is shown that when the traffic load is light, the delays for the VBR2 traffic and CBR traffic under unbalanced traffic condition are almost the same as those under the balanced traffic. When the load gets higher, the delays for VBR2 traffic and CBR traffic are a bit lower than those under the balanced traffic condition. From these facts, it has clearly shown that the M-WDMA+ protocol can dynamically allocate the network bandwidth according to the overall traffic load of the different types of traffic so that the performance of the network in terms of average delay could almost not be affected under the unbalanced traffic loads.

Fig. 15(b) depicts that the network under balanced and unbalanced traffic loads has almost the same overall throughputs when the traffic is not very high (normalized traffic load < 0.8). However, when the traffic load becomes heavier, the throughput of VBR2 traffic under unbalanced traffic load is lower than that of the VBR2 traffic under the balanced traffic load. It causes the overall throughput of the network under the unbalanced traffic load to be lower than that under the balanced traffic load. The reason of the lower throughput of the VBR2 traffic is clear. The amount of the VBR2 traffic under the unbalanced traffic condition with specified traffic ratio is quite smaller than that under the balanced traffic condition. The lower VBR2 traffic load results in the lower throughput of the VBR2 traffic. On the other hand, the overall throughput under the unbalanced traffic load has not decreased as much as that of VBR2 traffic because the network has allocated a portion of its bandwidth to serve the VBR1 traffic as it has increased its amount under the unbalanced traffic load. Due to the nature of the CNT subprotocol, the overall throughput under the unbalanced traffic load cannot be kept as high as that under the balanced traffic condition. This fact shows that under the unbalanced traffic load, the performance of the proposed protocol will only suffer a bit in terms of throughput. However, this degradation on the overall throughput is not serious, especially, when the total traffic load is not heavy.

VI. CONCLUSION

This paper introduced a new approach that combines different types of MAC protocols into a single WDM network to better serve a wide variety of multimedia applications. Some of the goals of this approach are: 1) to keep the advantages of the individual MAC protocols with respect to specific types of traffic streams; 2) to efficiently support a large range of traffic streams with different characteristics and QoS requirements in a single WDM network and for a transmitter to access the channel; and 3) to dynamically allocate the network bandwidth to the different classes of traffic in order to boost the network performance. We have investigated the performance of our M-WDMA network, and it was clearly shown that it outperforms state-of-art MAC protocols for the WDM networks in serving multimedia applications. As a result, we can reasonably expect the integration of various subprotocols into a single protocol to have a good potential for meeting the QoS requirements of future integrated services WDM networks.

REFERENCES

- T. E. Anderson, S. S. Owicki, J. B. Saxe, and C. P. Thacker, "High-speed switch scheduling for local-area networks," *ACM Trans. Comput. Syst.*, vol. 11, no. 4, pp. 319–352, Nov. 1993.
- [2] A. Azizgolu, R. A. Barry, and A. Mokhtar, "Impact of tuning delay on the performance of bandwidth-limited optical broadcast networks with uniform traffic," *IEEE J. Select. Areas Commun.*, vol. 14, pp. 935–944, June 1996.
- [3] K. Bogineni, K. M. Sivalingam, and P. W. Dowd, "Pre-allocation media access control protocols for multiple access WDM photonic networks," *Comput. Commun. Rev (USA)*, vol. 22, no. 4, pp. 235–46, Aug. 1992.
- [4] —, "Low-complexity multiple access protocols for wave-length-division multiplexed photonic networks," *IEEE J. Select. Areas Commun*, vol. 11, pp. 590–604, May 1993.
- [5] M. S. Borella and B. Mukherjee, "Efficient scheduling of nonuniform packet traffic in a WDM/TDM local light-wave network with arbitrary transceiver tuning latencies," *IEEE J. Select. Areas Commun.*, vol. 14, pp. 923–934, June 1996.
- [6] C.-S. Chang and K.-C. Chen, "Guaranteed quality-of-service wireless access to ATM networks," *IEEE J. Select. Areas Commun.*, vol. 15, pp. 106–118, Jan. 1997.
- [7] M. Chen, N. R. Dino, and R. Ramaswami, "Media-access protocol for packet-switched wavelength division multi-access metropolitan area networks," *IEEE J. Select. Areas Commun.*, vol. 8, pp. 1048–1057, Aug. 1990.
- [8] E. de Souza e Silva, H. R. Gail, and R. R. Muntz, "Polling systems with server timeouts and their application to token passing networks," *IEEE/ACM Trans. Networking*, vol. 3, pp. 560–575, Oct.r 1995.
- [9] W. P. G. and H. Levy, "Performance analysis of transaction driven computer systems via queueing analysis of polling models," *IEEE Trans. Comput.*, vol. 41, pp. 455–465, Apr. 1992.
- [10] M. Hamdi, "ORMA: A high-performance MAC protocol for fiber-optics LANs/MANs," *IEEE Commun. Mag.*, vol. 35, pp. 110–119, Mar. 1997.
- [11] M. Hamdi and W. C. Ming, "Performance evaluation of multiple access protocols using importance sampling," *IEEE/ACM Trans. Networking*, to be published.
- [12] P. A. Humblet, R. Ramaswami, and K. N. Sivarajan, "An efficient communication protocol for high-speed packet-switched multichannel networks," *IEEE J. Select. Areas Commun.*, vol. 11, pp. 568–578, May 1993.
- [13] F. Jia, B. Mukherjee, and J. Iness, "Scheduling variable-length messages in a single-hop multichannel local lightwave network," *IEEE/ACM Trans. Networking*, vol. 3, pp. 477–488, Aug. 1995.
- [14] A. E. Kamal and B. W. Abeysundara, "Hybrid medium access protocols for high-speed dual-bus fiber-optic local area networks," *Comput. Networks ISDN Syst.*, vol. 26, no. 5, pp. 563–584, Jan. 1994.
- [15] M. Kovacevic, M. Gerla, and J. Bannister, "Time and wavelength division multi-access with acousto-optic tunable filters," *J. Fiber Integrated Optics*, vol. 12, no. 2, pp. 113–132, 1993.
- [16] D. A. Levine and I. F. Akyildiz, "PROTON: A media access control protocol for optical networks with star topology," *IEEE/ACM Trans. Networking*, vol. 3, pp. 158–167, Apr. 1995.
- [17] B. Mukherjee, "WDM-based local light-wave networks Part I: Single-hop systems," *IEEE Network*, vol. 6, pp. 12–27, May 1992.
- [18] —, "WDM-based local lightwave networks Part II: Multi-hop systems," *IEEE Network*, vol. 6, pp. 20–32, July 1992.
- [19] G. N. Rouskas and M. H. Ammar, "Analysis and optimization of transmission schedules for single-hop WDM networks," *IEEE/ACM Trans. Networking*, vol. 3, pp. 211–221, Apr. 1995.

- [20] G. N. Rouskas and V. Sivaraman, "Packet scheduling in broadcast WDM networks with arbitrary transceiver tuning latencies," *IEEE/ACM Trans. Networking*, vol. 5, pp. 359–370, June 1997.
- [21] T. D. Todd, "The token grid network," *IEEE/ACM Trans. Networking*, vol. 2, pp. 279–287, June 1994.
- [22] P. Tran-Gia and R. Dittmann, "A discrete-time analysis of the cyclic reservation multiple access protocol," *Perform. Eval.*, vol. 16, pp. 185–200, 1992.
- [23] S. Tridandapani, J. S. Meditch, and A. K. Somani, "The MaTPi protocol: Masking tuning times through pipelining in WDM optical networks," in *IEEE INFOCOM '94*, vol. 3, June 1994, pp. 1528–1535.
- [24] J. C. Wu, A. E. Willner, and J. A. Silvester, "Protocols to eliminate tuning penalties for packet-switched WDM star networks with large tuning latency," *J. High Speed Networks*, vol. 6, no. 1, pp. 15–31, 1997.
- [25] A. Yan, A. Ganz, and C. M. Krishna, "A distributed adaptive protocol providing real-time services on WDM-Based LANs," J. Lightwave Technol., vol. 14, pp. 1245–1254, June 1996.
- [26] K. Bengi and H. R. van As, "CONRAD: A novel medium access control protocol for WDM local lightwave networks enabling efficient convergence of real-time and data services," in *Proc. Local Computer Networks*'2001, 2001, pp. 468–476.



Lixin Wang received the B.E. and M.S. degrees in computer science and engineering from the Tianjin University, Tianjin, China, in 1983 and 1986, respectively, and the Ph.D. degree in computer science from The Hong Kong University of Science and Technology, Hong Kong, in 2001.

He was an Assistant Professor in the Department of Computer Engineering at Tianjin University from 1986 to 1993. Since 1999, he has been with Trumptech Ltd., Shenzen, China, where he is the Chief Technical Director. He has published more

than ten academic papers in the area of wavelength-division-multiplexing (WDM) optical networks and performance analysis of multimedia communication protocols on shared medium networks. His current research interests are performance analysis of computer networks, optical networks, and wireless networks, as well as other forms of shared medium networks.



Maode Ma received the B.E. degree in automatic control from Tsinghua University, Beijing, China, in 1982, the M.E. degree in computer engineering from Tianjin University, Tianjin University, in 1991, and the Ph.D. degree in computer science from The Hong Kong University of Science and Technology, Hong Kong, in 1999.

He started his professional career as an Engineer in the computer industry in 1982. Since 1991, he was an Assistant Professor in the Department of Computer Engineering at Tianjin University. In 2000, he joined

the school of electrical and electronic engineering at Nanyang Technological University, Singapore, as an Assistant Professor. He has published more than 20 academic papers in the area of wavelength-division-multiplexing (WDM) optical networks and other relevant fields. His current research interests are performance analysis of computer networks, optical networks, and wireless networks.



Mounir Hamdi (S'89–M'90) received the B.S. degree in computer engineering (with distinction) from the University of Louisiana, in 1985 and the M.S. and Ph.D. degrees in electrical engineering from the University of Pittsburgh in 1987 and 1991, respectively.

He has been a Faculty Member in the Department of Computer Science at The Hong Kong University of Science and Technology, Hong Kong, since 1991, where he is now Associate Professor of Computer Science and the Director of the Computer Engineering Programme, which that has approximately

350 undergraduate students. From 1999 to 2000, he held Visiting Professor positions at Stanford University, Stanford, CA, and the Swiss Federal Institute of Technology, Lausanne, Switzerland. His general areas of research are in high-speed packet switches/routers and all-optical networks, in which he has published more than 180 research publications, received numerous research grants, and supervised some 20 postgraduate students, and for which he has served as Consultant to various international companies. Currently, he is working on high-speed networks, including the design, analysis, scheduling, and management of high-speed switches/routers, wavelength-division-multiplexing (WDM) networks/switches, and wireless networks. He is currently leading a team that is designing one of the highest capacity chip sets for terabit switches/routers. This chip set is targeted toward a 256×256 OC-192 switch and includes a crossbar fabric chip, a scheduler/arbiter chip, and a traffic management chip. In addition to his commitment to research and professional service, he is also a dedicated teacher.

Dr. Hamdi is currently or was formerly on the Editorial Boards of IEEE TRANSACTIONS ON COMMUNICATIONS, IEEE COMMUNICATIONS MAGAZINE, Computer Networks, Wireless Communications and Mobile Computing, and Parallel Computing and has been on the program committees of more than 50 international conferences and workshops. He was a guest editor of IEEE COMMUNICATIONS MAGAZINE, the IEEE JOURNAL ON SELECTED AREAS OF COMMUNICATIONS, and Optical Networks Magazine and has chaired more than five international conferences and workshops, including the IEEE GLOBECOM/ICC Optical Networking Workshop, the IEEE ICC Highspeed Access Workshop, and the IEEE IPPS HiNets Workshop. He is the Chair of the IEEE Communications Society Technical Committee on Transmissions, Access and Optical Systems and Vice-Chair of the Optical Networking Technical Committee, as well as the ComSoc technical activities council. He received the Best Paper Award at the International Conference on Information and Networking in 1998 out of 152 papers He received the Best 10 Lecturers Award (an annual award in which students universitywide choose the recipient from among all university faculty) and the Distinguished Teaching Award from The Hong Kong University of Science and Technology. He is a Member of the Association for Computing Machinery (ACM).